Author’s response to reviews

Title: Safety and adequacy of percutaneous kidney biopsy performed by nephrology trainees

Authors:

Vittoria Esposito (vittoria.esposito@icsmaugeri.it)
Giulia Mazzon (giulia.mazzon01@universitadipavia.it)
Paola Baiardi (paola.baiardi@icsmaugeri.it)
Massimo Torreggiani (maxtorreggiani@hotmail.com)
Luca Semeraro (luca.semeraro@icsmaugeri.it)
Davide Catucci (davide.catucci01@universitadipavia.it)
Marco Colucci (marco.colucci89@gmail.com)
Alice Mariotto (alicemariotto@outlook.it)
Fabrizio Grosjean (fabrizio.grosjean@gmail.com)
Giacomo Bovio (giacomo.bovio@icsmaugeri.it)
Ciro Esposito (espositociro56@live.it)

Version: 2 Date: 23 Aug 2017

Author’s response to reviews:

August 23rd, 2017

Afsaneh Khetrapal
BMC Nephrology

Dear Dr. Khetrapal,

All the authors of the manuscript entitled Safety and adequacy of percutaneous kidney biopsy performed by nephrology trainees BNEP-D-17-00288R1 thank you for the kind letter.

Below we address the editorial requests:
1. Consent for publication

If your manuscript contains any individual person’s data in any form (including individual details, images or videos), consent to publish must be obtained from that person, or in the case of children, their parent or legal guardian. Please provide details of this consent in this section. If your manuscript does not contain any individual persons data, please state “Not applicable” in this section.

Not applicable

2. Acknowledgements

Please use this section to acknowledge anyone who contributed towards the article who does not meet the criteria for authorship including anyone who provided professional writing services or materials.

Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements section. If no acknowledgements are necessary then please state 'None'.

None

3. Manuscript structure

Please refer to our submission guidelines and ensure that you structure your manuscript with the appropriate headings and subheadings as is shown: https://bmcnephrol.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article

We have ensured that the our manuscript structure has appropriate headings and subheadings

4. Final manuscript file

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables and figures should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

We upload our manuscript as a single, final clean version.

5. Authors' contributions
The individual contributions of ALL authors to the manuscript should be specified in this section with each author referred to by unique initials to allow them to be easily identified. Guidance and criteria for authorship can be found here: http://www.biomedcentral.com/submissions/editorial-policies#authorship

The individual contributions of all authors to the manuscript as specified below:

V.E. made substantial contributions to conception and design, acquisition of data, analysis and interpretation of data and writing the draft of the manuscript, gave final approval of the version to be published

G.M. made substantial contributions to conception and design and acquisition and analysis of data, gave final approval of the version to be published

P.B. made substantial contributions to design of the study and to analysis and interpretation of data, gave final approval of the version to be published

M.T. made substantial contributions to conception and design and interpretation of data and revising the manuscript, gave final approval of the version to be published

L.S. made substantial contributions to conception and design and acquisition and analysis of data, gave final approval of the version to be published

D.C. made substantial contributions to conception and design and acquisition and analysis of data, gave final approval of the version to be published

M.C. made a substantial contribution in interpretation of data, writing the draft of the manuscript and revising the manuscript, gave final approval of the version to be published

A.M. made substantial contributions to conception and design and acquisition and analysis of data, gave final approval of the version to be published

F.G. made a substantial contribution in interpretation of data, writing the draft of the manuscript and revising the manuscript, made a substantial contribution in interpretation of data, writing the draft of the manuscript and revising the manuscript

G.B. made a substantial contribution in interpretation of data, writing the draft of the manuscript and revising the manuscript, gave final approval of the version to be published

C. E. made substantial contributions to conception and design, analysis and interpretation of data and writing the draft of the manuscript, gave final approval of the version to be published

Best regards
Ciro Esposito